Author's response to reviews

Title: QuantiFERON-TB Gold In-Tube Implementation for Latent Tuberculosis Diagnosis in a Public Health Clinic: A Cost-Effectiveness Analysis

Authors:

Maunank Shah (mshah28@jhmi.edu)
Kathryn Miele (kmiele1@jhmi.edu)
Howard Choi (howard@jhmi.edu)
Susan Dorman (dsusan1@jhmi.edu)
Danielle DiPietro (ddipietr@tulane.edu)
Maria Martins-Evora (Maria.Martins-Evora@baltimorecity.gov)
Vincent Marsiglia (Vincent.Marsiglia@baltimorecity.gov)

Version: 2 Date: 5 September 2012

Author's response to reviews:

Dear Editor,

We apologize for any inadvertent prior overlapping text. We have reviewed the manuscript carefully and have attempted to removed all inappropriate overlapping text with any of our prior publications, particularly from the introduction.

Some of the overlapping text represented factually correct and independent statements in the current submission and prior manuscripts (i.e. description of some methods--"Model development and analysis utilized TreeAge Software"). Nonetheless, we have attempted to reframe these sentences to remove any overlapping text.

We hope the manuscript is now acceptable for peer-review.

Please let us know if any further revisions are needed.

Thank you,

Maunank